Author’s response to reviews

Title: Disease severity, debridement approach and timing of drug modify outcomes of adjunctive Azithromycin in non-surgical management of chronic periodontitis: a multivariate meta-analysis

Authors:

Nithya Jagannathan (nithyaj@hku.hk)
Aneesha Acharya (aneesha.a2@gmail.com)
Ong Farn (yifarn1984@yahoo.com)
Kar Li (skyli@hku.hk)
Luigi Nibali (lnibali@qmul.ac.uk)
Georgios Pelekos (george74@hku.hk)

Version: 3 Date: 12 Mar 2019

Author’s response to reviews:

Re: OHEA-D-18-00546R2

Disease severity, debridement approach and timing of drug modify outcomes of adjunctive Azithromycin in non-surgical management of chronic periodontitis: a multivariate meta-analysis

Nithya Jagannathan; Aneesha Acharya; Ong Yi Farn; Kar Yan Li; Luigi Nibali; George Pelekos

BMC Oral Health

Dear Editor,

Thank you for your review and in our responses to the comments as listed as follows:

Editor Comments:

Thank you for your revised manuscript.
Your manuscript is almost acceptable.

I found one syntax error.

95%ci

Please check the manuscript again.

We thank the editor for a detailed review. In the current revision have corrected the inconsistencies in reporting of the '95% CI' as suggested. In addition, we have also corrected the 'p value' outcomes now listed as 2 digit numerical throughout the draft and table 3 as earlier there were variable number of decimal digits listed.

We look forward to your further comments in this regard,

Best regards

George Pelekos